CS 39600: Metric Embeddings

Winter 2006 Harald Räcke

Preface

This course-material is based on the scribe notes of the course ”‘15-859(J): Metric Embeddings and Methods”’ that was given by Anupam Gupta and R. Ravi at Carnegie Mellon University, Pittsburgh in Fall 2003. I am most grateful to them for allowing me to use their material.

I am of course also grateful to the students who took that course and participated in the preparation of the scribe notes. So thanks to Luis von Ahn, Nikhil Bansal, Hubert Chan, Shuchi Chawla, Kedar Dhamdhere, Charlie Garrod, Daniel Golovin, Yiannis Koutis, Leonid Kontorovich, Peter Richter, Amitabh Sinha, Virginia Vassilevska, Shobha Venkataraman, and Ryan Williams.